Author's response to reviews

Title: Histopathological evaluation of ocular microsporidiosis by different stains

Authors:

Joveeta Joseph (joveeta@yahoo.com)
Geeta K Vemuganti (geeta@lvpei.org)
Prashant Garg (prashant@lvpei.org)
Savitri Sharma (savitri@lvpei.org)

Version: 7 Date: 22 June 2006

Author's response to reviews:

June 22, 2006

Chrissie Kouremenou
Assistant Editor
BMC - series journals
editorial@biomedcentral.com

Sub: MS: 6421603369057585 - Histopathological Evaluation OF Ocular Microsporidiosis by different stains

Dear Editor,
Thank you for considering our manuscript for publication in your esteemed journal. I am herewith submitting the revised manuscript entitled "Histopathological Evaluation OF Ocular Microsporidiosis by different stains" for publication in your journal. We have changed the format to the journal's style. The changes incorporated are:

Formatting changes requested
----------------------------------------
Affiliations - Each affiliation should be written in full and be provided
Reply - Done

Thanking you
Geeta K Vemuganti, MD
Ophthalmic Pathology Service
L.V.Prasad Eye Institute,
L.V.Prasad Marg, Banjara Hills,
Hyderabad - 500034, India
Email: geeta@lvpei.org